Author’s response to reviews

Title: Social Health Insurance Routine Data Analysis: The Incidence of Shortened Lifetime regarding Endoprosthetic Implants

Authors:

Roland Linder (prof.dr.roland.linder@wineg.de)
Hardy Müller (hardy.mueller@wineg.de)
Brigitte Grenz-Farenholtz (brigitte.grenz-farenholtz@wineg.de)
Caroline Wagner (dr.caroline.wagner@wineg.de)
Martin Stockheim (martin.stockheim@ruhr-uni-bochum.de)
Frank Verheyen (dr.frank.verheyen@wineg.de)

Version: 2 Date: 3 January 2012

Author’s response to reviews:

Dear Mrs. Calumpita:

Now I added the following ethics statement in the Methods section:

"Since all data have been anonymised and since exclusively aggregated statistics have been generated, there is no need to ask for an ethical approval."

With kind regards,

Roland Linder